Author's response to reviews

Title: The prevalence of pain and disability one year post fracture of the distal radius in a UK population: A cross sectional survey

Authors:

Catherine M Moore (catherine.moore@nottingham.ac.uk)
Jo Leonardi-Bee (Jo.Leonardi-Bee@nottingham.ac.uk)

Version: 6 Date: 22 September 2008

Author's response to reviews:

All revisions addressed as per the e-mail. Unfortunately I could not work out how to do the image formatting and wondered whether this could be done in house.